Author's response to reviews

Title: Nationwide shifts in the double burden of overweight and underweight in Vietnamese adults in 2000 - 2005: two national nutrition surveys.

Authors:

Ha T.P Do (dtpha@yahoo.com)
Edith J.M Feskens (edith.feskens@wur.nl)
Paul Deurenberg (paul.deurenberg@gmail.com)
Mai B. Le (bachmai_nin@yahoo.com)
Khan C. Nguyen (nguyencongkhan@vfa.gov.vn)
Frans J Kok (frans.kok@wur.nl)

Version: 6 Date: 22 January 2011

Author's response to reviews: see over
Dear Editor,

With this letter we would like to resubmit our finally revised manuscript entitled “Nationwide shifts in the double burden of overweight and underweight in Vietnamese adults in 2000 - 2005: two national nutrition surveys” (MS: 1440701337388786) for your journal, BMC Public Health upon your request of proof reading.

We have made several correction and revision in the text and in the figures as followings:

1. In the main part of manuscript, the following pages has been revised and addressed as tracked change:

2. In the figures” figure 3, 4 and 6 has been improved and uploaded.

We have read carefully the manuscript and tried our best to improve it as much as possible. We do expect that this final revised manuscript will be accepted for publication.

Thank you very much for giving our manuscript an opportunity to be reviewed and revised. I am looking forward to your acceptance.

With kind regards, and on behalf of the authors,

Dr. Ha T.P. Do
Head, Community Nutrition Department
Vietnam National Institute of Nutrition